Author's response to reviews

Title: Patient and Caregiver Perspectives of Health Provision Practices for First Nations and Metis Women with Gestational Diabetes Mellitus Accessing Care in Winnipeg, Manitoba

Authors:

Hannah Tait Neufeld (hneufel2@uwo.ca)

Version: 6 Date: 17 September 2014

Author's response to reviews: see over
BMC Health Services Research
Biomed Central
236 Gray’s Inn Road
London
WC1X 8HB
United Kingdom

September 17, 2014

Dear Editors:

This cover letter accompanies the revised research article manuscript I have prepared for your journal. As requested, I have addressed two final editorial requests and made the following changes to the manuscript document:

• A statement is included at the end of the Methods section on page 10 was added to confirm the Biomed Central RATS guidelines for reporting qualitative studies were adhered to
• Author information and competing interests sections were moved from the title page to the end of the conclusions on page 29
• Funding information was moved to the 'Acknowledgements' section on page 29

Thank you for your continued consideration and acceptance of the manuscript. Please let me know if you require any additional information or if I can be of any further assistance.

Sincerely,

Hannah Tait Neufeld, PhD
Banting Postdoctoral Research Fellow